**Author's response to reviews**

**Title:** Canadian infants' nutrient intakes from complementary foods during the first year of life

**Authors:**

James K Friel (frieli@cc.umanitoba.ca)  
Rhona M Hanning (rhanning@uwaterloo.ca)  
Corinne A Isaak (corinne.isaak@mts.net)  
Daniel Prowse (vprowse@foodfocus.com)  
Angela C Miller (angiecmiller@hotmail.com)

**Version:** 6 **Date:** 17 June 2010

**Author's response to reviews:**

Thank you for the revision request. All reference websites have been updated and verified. Tables have been revised as requested with titles placed outside the table. Please advise if further revisions are requested.